Author's response to reviews

Title: Detecting 22q11.2 deletion in Chinese children with conotruncal heart defects and single nucleotide polymorphisms in the haploid TBX1 locus

Authors:

Yue-juan Xu (xuyj1006@yahoo.com.cn)
Jian Wang (labwangjian@126.com)
Rang Xu (rang_xu@hotmail.com)
Peng-Jun Zhao (pjunzhao@sina.com)
Xi-Ke Wang (wangxike2008@sina.com)
Heng-Juan Sun (sunhengjuan@yahoo.com.cn)
Li-Ming Bao (lbaogene@yahoo.com)
Jie Shen (shenxiaomin@yahoo.com)
Qi-Hua Fu (qihuafu@hotmail.com)
Fen Li (lifen_88@yahoo.com.cn)
Kun Sun (sunkunyh@yahoo.com.cn)

Version: 5 Date: 23 November 2011

Author's response to reviews: see over
November 23, 2011

Dear Editor,

We appreciate all of the enlightening and helpful comments from the reviewers. The manuscript has been revised accordingly, and the modified parts are highlighted in blue. We have also responded to each reviewer’s comments point-by-point as listed below.

Thank you very much for your consideration of our manuscript.

Best regards

Yours sincerely,

Corresponding author: Kun Sun, Ph.D, Professor
Department of cardiology, Xinhua Hospital
Shanghai Jiaotong University School of Medicine,
1665 KongJiang Road,
Shanghai 200092,
P.R China

Response to reviewer's comments:
Reviewer: Dana Crawford
Comments to the Author
Table 3. There’s a minor typo. “3,427” as a chi-square value should probably be “3.427”. This is found twice and should be changed.

Author’s response:
Thank you for pointing this out. We have made changes accordingly in the revised manuscript. Please see the Table 3.